A CORRELATIONAL STUDY OF READING SPEED PERFOS AND THE EXTENT OF VOCABULARY AMONG STUDENTS OF PESANTREN IMMIM UJUNG PANDANG (A CASE STUDY)



Presented to the Faculty of Letters

Hasanuddin University Ujung Pandang in Partial

Fulfilment of the Requirement for the Degree of

Sarjana Sastra

BY

MUNATZIR

Reg. Numb. ; 85 07 265

UJUNG PANDANG, SEPTEMBER 1991

UNIVERSITAS HASANUDDIN FAKULTAS SASTRA KAMPUS BARU TAMALANREA UJUNG PANDANG

Sesuai dengan surat tugas Dekan Pakultas sastra Universitas Hasanuddin nomor : 3562/PTO4.H5.FS/C/1990 tertanggal 31 Desember 1990, dengan ini kami menyatakan menerima dan menyetujui sekripsi ini.

UP. 1991

Pembimbing Utama

(Drs. John Ruru, M.A.)

Pembantu Pembimbing

(Dra. Etty Bazergap, M.Ed

Disetujui untuk diteruskan kepada panitia ujian sekripsi

Dekan

Ub. Ketua Jurusan

(Drs. Mustafa Makka, M.A.)

UNIVERSITAS HASANUDDIN FAKULTAS SASTRA KAMPUS BARU TAMALANREA UJUNG PANDANG

Sesuai dengan surat tugas Dekan Fakultas sastra Universitas Hasanuddin nomor : 3562/PTO4.H5.FS/C/1990 tertanggal 31 Desember 1990, dengan ini kami menyatakan menerima dan menyetujui sekripsi ini.

UP. 1991

Pembimbing Utama

(Drs. J.H. Ruru, M.A.)

Pembantu Pembimbing

(Dra. Etty Bazergap, M.Ed)

Disetujui untuk diteruskan kepada panitia ujian sekripsi

Dekan

Ub. Ketua Jurusan

(Drs. Mustafa Makka, M.A.)

UNIVERSITAS HASANUDDIN FAKULTAS SASTRA

Pada hari ini, Sabtu tanggal 28 September 1991
Panitia Ujian Sekripsi menerima dengan baik sekripsi
yang berjudul; A CORRELATIONAL STUDY OF READING SPEED
AND THE EXTENT OF VOCABULARY AMONG STUDENTS OF PESANTREN IMMIM UJUNG PANDANG (A CASE STUDY) ini yang diajukan dalam rangka memenuhi salah satu syarat guna
memperoleh gelar sarjana Sastra jurusan Sastra Inggris
pada Fakultas Sastra Universitas Hasanuddin Ujung Pandang dengan susunan panitia penguji sebagai perikuty:

- 1. Hamzah A. Machmoed, M.A. Ketua
 - Sekretaris

- 2. Drs. Alwy Rahman
- 3. Prof. Dr. R. Soewondo, M.A. Anggota
- 4. Dra. Badrah Hussein, M.S. Anggota
- 5. Drs. J.H. Ruru, M.A. Anggota
- 6. Dra. Etty Bazergan, M.Ed. Anggota

. Et Janzerge

ACKNOWLEDGEMENT.

"BISMILLAHIR RAHMANIR RAHIIM" the writer would like to express his gratitude to ALLAH SWT., the Almighty. Because only of his will the writer could finish in writing this thesis. The thesis would have never been completed without the assistance of my classmates, and guidance of my academic advisers and consultans. Thus the writer would like to thank to the following individuals:

- Dr. Najamuddin, M.S., the Dean of the Faculty of Letters Hasanuddin University Ujung Pandang.
- Drs. Mustafa Makka, M.S., the Head of English Departement of the Faculty' of Letters Hasanuddin University Ujung Pandang.
- 3. Drs. J.H. Ruru, M.A. and Dra. Etty Bazergan, M. Ed. the writer's first and second consultan for their help and advise in writing and finishing this thesis. The writer would not have been able to complete this thesis without both of them.
- 4. Drs. Agustinus Ruruk L., M.A. and Drs. Muh. Syafri Badaruddin, the writer's academic advisers who always give the witer advise during his studies at the English Departement of the Faculty of Letters Hasanuddin University Ujung Pandang.
- 5. Dra. S.A.A. Ruru, M.A., Drs. M.L. Manda, M.A., Drs. O.J. Wehantouw, M.S., Dra. Herawaty, Drs. Ibnu Nan-

- 6. Drs. Azhar Arsyad, M.A., the Readmaster of the Pondok Pesantren Putra IMMIM Ujung Pandang and his students who have given opportunity to the writer to take data at their Pondok Pesantren.
- 7. Drs. Fahmi Room, Drs. Syamsulbahri, Munawar S.H., As'ad Kadir's family, Ir. Usman Kadir's family, Yacob, Ihsan, Laode, Rusli, Hasanuddin, Ismail, Yati, Lucy, Paula, Fahmi, Batariana, Irma, and still many others of the writer's friends that could not be mentioned one by one who always give the contribution of thinking to him during his studies.
- 8. Finally, the writer whishes to offer his gratitude with all his heart to his father, Fachruddin Monro, B.A., and to his mother, St. Halwiyah R., who are never tired to give their love and kindness and many other things that cannot be mentioned until today, and so to all his lovely little brothers and sister, Ahmad Iqbal Fachry, Faridha Fachry, Ahmad Farid Fachry, Ahmad Jamil Fachry, and to his first cousin, Ratnawati, Ir. Maknun, Nuraeni, and to his nephew, Drs. Lukman Hakim and to lovely Suhartini for their help, encouragement, and assistance. With out them this thesis would have never been in existance. And Allah Knows Best.

TABLE OF CONTENTS

				4	- *:					(3)	- 3	+;									1	PAGE
TITL	Е			220						21		5.5	0.22		1025		Je.	272		20.		i
		APPR																				
		DGEM																				iv
																						000020
		CON																				
LIST	OF	TABL	ES .	٠.,	٠.	• • •	٠.,	٠.	٠.	٠.	•	• •	• •	٠.	٠.	٠	٠.	٠.	•	• •	•	wii:
LIST	OF	APPE	NDIC	ES	٠.				٠.					٠.	٠.	•	٠.	٠.	٠	٠.		x
ABST	RACI				٠.								٠.	٠.				٠.				xi
I.	INTE	ODUC	TION																			
	1.2.	A	son pe o hodo .1.	for f t log Met	he y ho	Pr and	sir cok d F	ole Pro Co	m ce	edu	Ti	it e in	le g	Da	ta		::	::	:			3 4 4 4
		1.4	.1.1 .1.2 .2.1	. L	ho	rai	cy of	Re	semp	os	ch	n ng	· ·	 he		h	es	is		• •	:	4
		1.4 1.4 1.4	.2.2 .3.	Pro	ce	sor dur	a's ce eti	of on	ro	ol	10	tcl	Mo ti	me ng	nt	a	ta	al 	у:	si 	s	5 5 5
	1 5	1.4	.3.2 .3.3	. I	ns:	trı iqı	ime	of	A	na	i	· · ·	in	g	th	· ie	 D	··	a	• :	:	8
		Seq																				
II.	REVI	EW O	FRE	LAT	ED	L	TE	RA	TU	RE												
	2.2.	c.	els	of ia:l ent rme	Rea Lary dia	eve y I	ing	rel	el	:	:		::	::	::	:		::	:	::	:	14
	2.3.	Read a. I b. I c. I	ding Eye Retu Vary The	Spa Spa rn ing Tha	eed n Eye ti ugi	d .	Ra	ep		::				::				::	:	::		17 19 19 20 21
	2.4.	2.4	Recombul.	ary Con Wor	te:	nr: xt Ana	Cl Cl	me ue si	nt s s	::	:	:	::	::	::	•	::	::	:	: :	:	22 26 29 31
	1000		Ext																			40

III.	DATA	ANALYSIS AND DISCUSSION	
3.5	3.1.	Presentation of Data Analysis 44 3.1.1. The Analysis of Data Collected	4
		Through Reading Speed Tests 4	4
		3.1.2. The Analysis of Data Collected	•
		Through Vocabulary Enrichment test 46	6
		3.1.3. The Analysis of Data Correlation . 50	6
		3.1.4. The Analysis of Data Collected	
		Through Questionnaire 59	9
	3.2.	Discussion of Data Analysis 70	0
IV.	CONC	LUSION AND SUGGESTION .	
	#: i:	Conclusion 76	5
		Suggestion	
	4	suggestion	•
BIBL	IOGRAI	PHY 7.0	8
APPEN	NDIX .		0

1

LIST OF TABLES

			PAGE
* *			
Table	1.	The standard of correlation product moment	10
Table	2.	The standard of Reading Speed of the whole types of reading material	42
Table	3.	The raw result of the students' reading speed in word per minute	44
Table	4.	The raw data of the students' vocabula- ry enrichment	47
Table	5.	The students average value of reading speed and vocabulary enrichment	49
Table	6.	The mean score of reading speed	52
Table	7.	The mean score of vocabulary enrichment	53
Table	8.	The presentation of the correlation product moment analysis	54
Table	9.	Preparation table in looking for the correlation between reading speed and the extent of vocabulary	57
Table	10.	Are you interested in learning English?	60
Table	11.	Do you often or ever found information about reading speed and vocabulary en - richment?	61
Table	12.	What is your opinion about speed read - ing and the extent of vocabulary ?	61
Table	13.	Are you interested in improving your reading speed and vocabulary ?	62
Table	14.	Do you believe that reading speed and vocabulary can be improved ?	63
Table	15.	Do you often read an English book, news paper, magazine, or novel with a purpose to improve your reading speed and enrich your vocabulary	63
Table	16.	What is the average lenght of time do you spend for reading in everyday	64

Table	17.	How often do you go to the library for reading in a week ?	64
Table	18.	Before starting to read, do you often realize your reading purpose ?	65
Table	19.	Do you often do regression in your reading ?	66
Table	20.	When you are reading, do you know what do you read ?	66
Table	21.	Do you often practice your speed read- ing and enrich your vocabulary well ?	67
Table	22.	When your teacher explains the lesson in English, do you understand what is he explaining ?	67
Table	23.	What is your effort to increase your reading speed and enlarge your vocabulary ?	68
Table	24.	What factors hamper you in doing speed reading and enlarging your vocabulary?	69

LIST OF APPENDICES

			PAGE
Appendix,	Α.	The table of the critical value of (r) product moment	80
Appendix	в.	Reading Speed Test	81
Appendix	c.	Vocabulary Enrichment Test	88
		Questionnaire	

ABSTRACT

Dengan semakin pesatnya kemajuan Ilmu Pengetahuan dan technologi sebagai konsekwensi nalar serta kemampuan daya pikir umat manusia memberi dampak terhadap pola kehidupan masyarakat, baik itu secara langsung maupun tidak langsung. Hal ini adalah merupakan tantangan yang harus kita hadapi serta wajib kita pecahkan bersama. Dengan kata lain kita ditantang untuk mengadaptasikan diri terhadap perkembangan Ilmu pengetahuan dan technologi tersebut.

Ada banyak cara yang tersedia untuk menjawab tantangan era modernisasi ini. Salah satu diantaranya adalah dengan banyak membaca buku-buku untuk memperoleh informasi tentang Ilmu pengetahuan dan technologi yang kita butuhkan.

Dengan membaca kita akan memperoleh banyak pengetahuan tentang budaya antar bangsa. Pengetahuan tentang sejarah dan budaya dari orang-orang sebelum kita yang erat kaitannya dengan kejadian-kejadian masa kini dan bahkan dengan peristiwa-peristiwa yang akan terjadi pada masa yang akan datang.

Membaca adalah suatu hal yang sangat penting bagi semua orang, khususnya bagi siswa-siswi yang ingin mem perluas wawasan berfikir dan pengetahuannya yang telah diperolehnya baik didalam kelas maupun diluar sekolah. Terlebih lagi membaca dapat pula meningkatkan harkat dan martabat kita sebagai mahluk yang berbudaya.

Sekripsi ini berjudul "A Correlational Study of Reading Speed and The Extent of Vocabulary Among Students of Pesantren IMMIM Ujung Pandang; yaitu suatu study kasus yang mencoba menemukan hubungan timbal balik antara kecepatan membaca dan perluasan kosa kata yang dimiliki oleh santri-santri (siswa-siswa) kelas tiga Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang.

Sehubungan dengan hipotesa bahwa terdapat hubungan timbal balik antara kecepatan membaca dengan perluasan kosa kata, maka penulis telah melakukan suatu penelitian mengenai kedua variabel tersebut diatas.

Data yang diperoleh dalam penelitian ini dianalisa dengan mempergunakan Pearson's Product Moment Analysis yang kemudian membuktikan adanya korelasi positif antara kecepatan membaca dan perluasan kosa kata
pada santri-santri dari Pondok Pesantren tersebut. Dan
sekalipun kecepatan membaca mereka rata-rata masih ter
golong rendah, namun penguasaan dan perluasan kosa kata yang mereka miliki sudah dapat dikatakan cukup baik,

Setelah melalui beberapa rintangan yang cukup be-

rat, penulis akhirnya berhasil memperoleh tingkat korelasi antara kecepatan membaca dan perluasan kosa kata senilai 0,530 dari santri-santri semester enam di
Pondok Pesantren itu. Dan setelah tingkat nilai tersebut dicocokkan dengan tabel nilai-nilai kritik (r), di
peroleh angka 0,463. Hal ini berarti bahwa kedua variabel tersebut (perluasan kosa kata dan kecepatan membaca) memiliki tingkat korelasi yang positif. Posisi
signifikansinya berada antara 0,400 sampai 0,600 yang
dapat dilihat pada tabel 1 dalam sekripsi ini.



CHAPTER I

1.1. Background of Writing

Learning English as a foreign or second language covers four skills, they are listening,
speaking, reading, and writing. Among the four
skills, reading gets greater attention than three
other ones. The fact shows us that most people
prepare themselves to be able to read as well as
to understand the English reading materials. This
is because many of the scientific books are written in English.

Alwasilah (1985:163) says that according to UNESCO at least 62% of the scientific books are written in English while the people who can read and understand English as a foreign or second language is still very limited. Probably most of them are only the students of English Department in colleges and courses. This number does not guarantee that they can obtain information much. This is because they do not have capability to read fast. And if we compare with the issues of information which are published in every day of course the number of the reader is too small.

In the Pondok Pesantren Modern Pendidikam Al

learned from the first to the last semester. It is about six years, two semesters each year. The English subject is broken down into structure, or tenses, speaking, wocabulary building, etc. In the reading subject the students are taught to seek information from the reading materials which consist of analyzing the main idea, supporting detail, and the way the writer expresses his idea as well. Moreover they are taught how to improve their reading skills. Nevertheless it is not tought as detailed as found in college curriculum.

Some of the researchers of English have found the data that most of the readers of English as a foreign or second language still could not read well. And this is responded seriously by many teachers and students who are concerned with the reading problems. Rajab Johary in his note (1986) says that the average of the students still could not read the English text book better and effectively. This causes most of them feel difficult to read English books. Moreover, this is affecting the inability of the students to read rapidly. This statement are supported by Cristine and Smith (1988; 45) who say that the average reading for the English as a foreign or second language students is approximately 120 to 150 words per minute. While

according to Celce and his friends (1979:35) the the average of them have to reach at least 200 words per minute.

1.2. Reason for Choosing the Title

Realizing the importance of reading because of the development of science and technology, the writer is interested in writing one of the aspects of reading which is needed today. This is known as 'reading speed' and 'vocabulary enlargement'.

Thus the main reason why the writer is interested in writing about the reading speed and the extent
of the vocabulary is because of the development of
the science and technology which demand people to
hunt information as much as possible. It proves
that people have to be a good reader, or have to
be a good speed reader.

That the Minister of Education and Culture of the Republic of Indonesia, Rifai (1979:v), in one of his decicion letter states that the aim of English teaching in the secondary schools is to give the pulpils a 'working knowledge' of English which is useable as one of the tools in understanding lectures and text books in the university later on.

Another reason is to fulfill one of the rules in finishing the study at the English Departement of Hasanuddin University, that is by writing a scientific work.

1.3. Scope of the Problems

The scope of this research is to find out and describe the correlation between reading speed and the extent of vocabulary doing by the students of the Pondok Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang. The research is limited to the third year students only as those students are the ones who have learned English for three years, or six semesters.

1.4. Methodology and Procedure

1.4.1. Method of Collecting Data

1.4.1.1. Library Research Library research was applied for getting some information from the text-books which have relationship to the topic of the thesis.

1.4.1.2. Field Research

Field research means that in collecting the data, the writer presents written tests such as read ing speed test, vocabulary enlarge
ment test and questionnaire. There
was an interview also in order to
get more information about those
samples.

1.4.2. Method of Composing the Thesis

1.4.2. Method of Composing the Thesis

- 1.4.2.1. Descriptive Method

 The writer uses the descriptive method to describe the data which are obtained through the thesis.
- 1.4.2.2. Person's Product Moment Analysis

 Person's product moment analysis

 is the technique used to show the

 correlation between the two vari
 ables; reading speed and the ex
 tent of the vocabulary.

 Sutrisno Hadi (1987:273) says :

 "korelasi produk moment melukiskan

 hubungan antara dua gejala inter
 val... gejala interval adalah ge
 jala yang menggunakan skala peng
 ukuran yang berjarak sama".

1.4.3. Procedure of Collecting Data

1.4.3.1. Population

The population of the research is the students of Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang. These students are choseen as the population since they are considered to have passed the English subject which they studied

six semesters for three years; two semesters each year.

1.4.3.2. Sample

The sample of the research is the third year students of the Pondok Pesantren IMMIM Ujung Pandang. The sampling technique of this research is total sampling.

1.4.3.3. Instrument

The instrument of the research consists of two reading speed tests, two vocabulary enlagement tests, and a questionnaire. The description of these instruments is as follows:

Reading Speed Test

The reading speed test used in this research consist of two, namely:
One Night On The Mt. Everest Of
Man-Made Structure (520 words)
and Thanksgiving Day (600 words).
These tests are used to measure the reading speed. They are conformed to be read by the population of the research who are in pre-advanced and in advanced levels. This is

supported by Heaton (1975) in La Uhudu (1989:15) who says that advanced level of reading material consists of 400 to 600 running words.

Vocabulary Enlargement Test

Like the reading speed tests, Vocabulary enlagement test also comprise three kinds. These tests are
taken from the reading speed tests.

Each of the vocabulary enlagement
test of the two reading materials
has fifteen numbers. Each of them
values one score. It means that if
all the answers of the students are
correct, they get 100 %.

Questionnaire

The questionnaire is used to collect information about the students backround in learning English; their interest in reading subject in general and in speed reading in particular.

The questionnaire aims to get information of data of the students' abilaty to read speedily; whether the

students of the Pondok Pesantren of the third year able to do speed reading and enriching vocabulary or not.

The writer collects the data through written tests. In collecting the data, the writer distributes the written instruments to the third year students of the Pondok Pesantren. Those tests were carried out on Monday 10, June 1991. After distributing the tests, the writer took a few minutes to give some explanations of how to work with those tests and then lets the students to do them. Each reading speed test should be read as quickly as possible. While the vocabulary extending tests should be done as well as the students' capabilities. After finishing the reading speed and the vocabulary extending tests, the questionnaire was distributed. The time used to complete the data distribution is about 115 minutes. That is ranging from eleven o'clock to five to one o'clock p.m.

1.4.4. Technique of Analyzing Data

For the first time the writer analyze the data of reading speed tests by using the formula as follow;

have been read . X 60 = WPM*
number of time
used (second
elapsed)

Secondly, the writer analyze the data of vocabulary extending tests by using the percentage system as follows;

total number of question's items

X 100 = VE*

The words in the both formulas stand for; WPM = Words Per Minute

VE = Vocabulary Enrichment

To make the researcher easier to analyze in the correlation product moment analysis, the data are grouped from the highest to the lowest.

To know the degree of correlation between the reading speed and the extent of vocabulary, he used the formula of Pearson's product moment analysis as follows;

$$r_{xy} = \frac{\Sigma xy}{\sqrt{(\Sigma x^2)(\Sigma y^2)}}$$

The words in the formula stand for;

"xy = coeficient variable between variable
"x" and "y".

 $\sum xy = sigma product moment "x" and "y".$

To interpret the result of the correlation analysis, the writer uses the standard of correlation product moment (r) as in the following table;

Table 1

	(*)	r	interpretation
Between	0,800	to 1,000	High
Between	0,600	to 0,800	Substantial
Between	0,400	to 0,600	Moderate
Between	0,200	to 0,400	Low
Between	0,000	to 0,200	Very low (Neglible or no . correlation)

• To know the significance of correlation between reading speed and the extent of vocabulary, the writer uses the table of the critical value of (r) product moment which was composed by L.D. Edmison of (r) as follows;

$$r = \sqrt{\frac{t^2}{n - 2 + t^2}}$$

The complete table of this formula is attached in the Appendix of this thesis.

The function of the table shows us the interval of the significance based on the po-



pulation of the sample.

The last data analyzis is the result of the questionnaire. The data acquired from the questionnaire were tabulated into a table of frequency, and each of them was analyzed individually. All of the findings can be seen completely in the chapter of presentation of data.

1.5. Objective Of The Research

The research aims at finding out the correlation between reading speed and the extent of the vocabulary of the third year students of Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang.

The result of the research is expected to be used by whoever interested in improving his skill of reading speed and vocabulary enriching, especially for the students of the Pondok.

1.6. Sequence Of Presentation

The Thesis will be composed in four chapters; Chapter One includes the Introduction that is describes the backround of writing, reasons for choosing the title, the scope of the problems, methodology and procedure, objective of the research, and the sequence of presentation.

Chapter Two deals with review of related literature which describes some concepts of reading such as;

definition of reading, levels of reading, reading speed, vocabulary enrichment, and the correlation between reading speed and the extent of the vocabulary.

Chapter Three deals with the data analysis and the discussion of the reading speed tests, vocabulary enriching tests, questionnaire, the correlation between reading speed and the extent of vocabulary, and the result of the research.

Chapter Four deals with the conclusion of the research and will present some suggestions from the writer's point of view.

CHAPTER II

REVIEW OF RELATED LITERATURE

2.1. Definition of Reading

Reading has various definition as the variety of people's point of view.

Terry and Thomas (1977:284) define reading " as the perception of the written symbols as meaningful, involving recognition of words, fluency, and comprehension ".

Reinking and Sceiner in Kustaryo (1988:2) said that " reading is instantanious recognition of various written symbols with exiting knowledge and comprehension of the information and ideas communicated ".

Hornby in his Oxford Advanced Learner's Dictionary of Current English (1974:698) said that " reading is the way in which something interpreted or understood ".

Based on the definitions above the writer then comes to a conclusion that reading is the combination of some components that result in one action that is known as reading. This statement is supported by Smith (1973:6) who says that " reading involves two aspects of the reader himself, that is internal and external aspects ".

In addition, Nurhadi (1987:13) says that

"reading is a complex and complicated process. It involves the internal and external factors of the readers. The internal factor means everything which comes from the reader that can result in interaction in the reading process. While the external factor means everything which has relationship with the reading material and the environment where the reading is taking place".

Further more, Nurhadi classifies the internal and external factors as follows:

Internal factors include; intelligence, interest, attitude, talent, motivation, and the aim of reading.

External factors include; the reading text, environtment, social economic backround, habit, and the tradition of reading.

"the activity that consist of the combination of words recognition, intellect, and emotion interrelated with the prior knowledge to understand the massage being communicated".

2.2. Levels of Reading

In learning languages, reading is one of the language skills that should be developed. And if we talk about the skill of reading, each person may have different capability. The skill of someone is determined by his creatifity and capability when he is reading. In relation to this, reading material also affect a reader in developing his reading skill. Based on this, psycholinguists, particulary those who are expert in reading, make classification of reading levels because not all

the readers have the same level. Simanjuntak
(1988:26) says that those experts set up the
levels into four as can be seen in the following
points:

a. Initial Level

The instructional goals in initial level are the students will be able to;

- read silently a passage of at least two paragraph in lenght
- demonstrate comprehension by responding to oral or written question
- understand written directions.

b. Elementary Level

The instructional goals in elementary level are that the students will be able to;

- read short English selection of passages,
 demonstrate and understand the content by
 locating specific information
- identify sequence of event
- recognize cause and effect and perceive organization.

The study skill of this level are the abilty to;

- read and follow instruction or direction
- use or consult reference material
- comprehend information on graphs, maps, charts, and table as well.

c. Intermediate Level

The instructional goals of this level are the students will be able to:

- continue to refine reading skills acquired at the earlier levels
- refine words attack skills
- develop critical reading ability
- apply reading ability to comprehend more complex content areas of the instructional materials
- scan for concluding sentences, main ideas,
 and specific information.

The study skills of this level are:

- reading and following instruction or direction
- skimming to locate information quickly
- evaluating information found in reverence materials.

d. Advanced Level

The instructional goals of this level are the students will be able to;

- continue to refine the reading skills acquired at earlier levels and will also refine text attack skills
- develop critical reading ability
- apply reading ability to the development of

the study skills

- read selection of increased difficulty for study purposes
- develop greater ability in comprehending more complex content are instructional material
- scan for concluding sentences, main ideas,
 and specific information.

The study skills in advanced level are;

- using content area of text books reference materials used by native speaker of English
- scanning to locate specific details of information and adjust the rate of speed to the level of difficulty of a selection.

After seeing the four levels of reading curriculum above, it can be concluded that reading
speed and the extend of vocabulary are included
in the whole levels of reading, especially, in
intermediate and advanced level.

2.3. Reading Speed

Many people are still confused about the term reading speed and speed reading. One of them is Tompubolon (1987:7) who stated that " speed reading and reading speed have no difference. The two can be used to state how many words a person reads in each minute ".

The statement above is not accepted by Terry

the study skills

- read selection of increased difficulty for study purposes
- develop greater ability in comprehending more complex content are instructional material
- scan for concluding sentences, main ideas,
 and specific information.

The study skills in advanced level are;

- using content area of text books reference materials used by native speaker of English
- scanning to locate specific details of information and adjust the rate of speed to the level of difficulty of a selection.

After seeing the four levels of reading curriculum above, it can be concluded that reading
speed and the extend of vocabulary are included
in the whole levels of reading, especially, in
intermediate and advanced level.

2.3. Reading Speed

Many people are still confused about the term reading speed and speed reading. One of them is Tompubolon (1987:7) who stated that " speed reading and reading speed have no difference. The two can be used to state how many words a person reads in each minute ".

The statement above is not accepted by Terry

and Thomas (1977:319). According to them " speed reading or rapid reading is the technique for improving the speed and comprehension in reading.". If so, reading speed and speed reading are different;. That is to say that speed reading is used to acquire reading speed, while reading speed is the total number of words were read in each minute.

There are so many people who stated that the ability of the reader of English as a foreign or second language to read speedily is low. Their reading speed is below the standard of 200 words per minute. Marianne (1979) says that " the reader of English as a foreign or second language have very little speed in reading. Most of them could not achieve 200 words per minute ".

In addition, Nuttle and Smith (1988:45)
say that " the average reading speed for the reader of English as a foreign or second language is
120 to 150 words per minute before training, and
after training it is hope that he or she could
read at about 200 words per minute ".

Thus it is clear now that the students of normal reading speed for the reader of English as a foreign or second language is 200 words per minute.

To obtain a reader's reading speed in each

minute, Soedarso suggests the following formula;

Number of words were read x 60 = WPM second elapsed Soedarso (1988:14).

In relation to the statement about the low reading speed of a reader of English as a foreign or second language, the writer tries to present ways that can be used to increase reading speed as follows:

a. Eye Span

Eye span is one of the eye movement development. In this eye movement, according to
Zorn (1980), the eyes of a reader should be
trained to read one word to two words, to four
words, to five words, to whole line at one time.
This is done to enable the reader to read phrases or large group of words.

b. Return Eye Sweep

Return eye sweep means the eyes of the readers move as they go from the end of one line to the beginning of the next line. It should be a very rapid movement, just as the words sweep implies. The eye should have a return which is smooth, rapid, and rhytmic.

In this method the eyes are determiner of the reader's speed reading. That is to say that if the eyes of the reader always waver from the end of one line to the beginning of the next lines, his speed reading suffer. And there is one thing the reader have to remember in doing this method, that is the forcefulness and direct movement are the keys of improving the return eye sweep method.

The following is an example of the using of the return eye sweep method which is quoted from Zorn (1980:57). Pay attention to the bow marks.

sweep your eyes down and back
to the next line never stop
at the end of one line
always force yourself to sweep
down and back down and back
this motion should be quick
and without hesitation.

In the return eye sweep the eyes should make two eye span or jump per each line of the paragraph. At the end of the line, the eyes follow the dotted arrows which show the readers the return eye sweep pattern.

c. Varying the Rate

Varying the rate is essential to speed reading. It can increase the reader's speed per

minute. The emphasis of this eyes movement is that the speed reading should be adjusted the type of reading text. This method is the same as the flexibility rate in reading. Soedarso (1986:18) stated that speed reading should be flexible. It means that speed should not always the same for all types of reading material. The speed should be adjusted to the types and purposes of reading.

In addition, Zorn (1980:92) says that "reading speed should be geared to different variables such as the purpose in reading the particular selection, the level of difficulty of the selection, the type and style of the author and text, and the familiarity of the material.

d. The Thought Unit

span method but with a twist. The idea here is to acquire the habit of reading one thought after another. Regardless of the number of word infolved. This require a lot of flexibility of the part of the reading because he has to really vary his eyes movement in order to accommodate the various length of the different thoughts. This technique stresses that all con-

sentration should be devoted to grasping the author's thought and ideas. The expansion of the eye span from one word to another words or so on is underplayed in favor of consentrating on the large blocks of thought presented in the writing.

e. Recognition Rate

Recognition rate also plays a very important role in increasing the speed reading. In rerelation to this Zorn (1980:85) said that
" recognition rate is one of the basic factors
that must be taken into account to become a
speed reader".

Most of the readers read the English texts with long recognition rate. We can see the fact that when the students are tested with their lessons, the average of them force themselves to read the theory or the material of the subject to be examined. But, however, when they are asked to answer questions, most of them could do nothing. It means that they have low recognition rate in their reading.

Thus slow recognition rate causes slow reading rate. It assume that the reader in this case is having difficulties in recognizing the words and this result in slow reading speed.

Knowing the technique of how to read speedily is not enough yet to make a slow reader to be a speed reader. He also need to know the factors that hamper him to read speedily. Simanjuntak (1988:46) pointed out that lip movement is one of the factors that hampers the reader to read rapidly. According to him, lip movement consists of two factors, they are vocalization and subvocalization. More information about them each, according to Simanjuntak, is as follows;

"Vocalization means a reader moves his lips or whisper each word as he reads, or he is vocalizing the word... and obviously the reader will never have a high reading speed"

Furthermore, he also pointed out that one thing has to be remembered is that in reading process, the eyes are really the servants of the minds. They should do the minds bidding in the reading process. So they see the printed words and at the same time the mind or the brain interprets them. Therefore when the reader vocalizes the words his eyes see, he will be slow in his reading. This happens because the capability of the brain to interprets the words is faster than the capability of the mouth to vocalize them.

At the same page, Simanjuntak also pointed out that " subvocalization means a reader does not whisper the words, but he says the words to himself. He does not consider himself reading if does not hear his voice uttering the symbols.

Soedarso (1989:5-8) describes six matters that hamper a reader in speed reading as follow:

Vocalization in reading hampers a reader to read faster because he must pronounce word by word completely voiced or voiceless.

b. Lip Movement

Most adult readers read the text by moving their lips; 1.e. they utter word by word with trembling lips. When a reader uses this technique in his reading, his speed will slow down. He will have the same speed as the vocalized reader.

c. Head Movement

A reader who uses this way in his reading always move his head as the eyes move from one line to the next lines. This way is utilized by children when they are reading because they still could not span their eyes widely. They are only able to see the next word from what they have read. They read from one word to the next word. This is also known by 'peripherial vision'.

d. Regression



d. Regression

One of the common habit that blocks the speed reading is called 'regression'. Regression is the back up of the eyes to read the passing words. Of course this way of reading will hamper the reader to read speedily.

e. Following the Moving Object

Following the moving object is using an object to help to read rapidly; i.e. using a finger to point the words while the eyes are following that object. This is a child's habit in reading. In this case, of course, the speed depends on the object movement and without the object the reading speed will decrease.

f. Subvocalization

A subvocalization reader reads by pronouncing the words in his mind. The reader is slow
down by waiting to hear the words mentally. We
should know that reading the words mentally
causes many readers not read as fast as they
want to.

As an addition, Zorn (1980:105) added two more kinds of matter that hamper a reader in speed reading besides those pointed out by Soedarso above. The two are as follows:

g. Lack of Drive

Lack of drive means that the readers do not have the recognition rate in his reading as they not possess motivation to read speedily. That is why they do not force themselves to read faster and break their old habit.

h. Overconcern With Comprehension

Overconcern with comprehension means that the reader emphazises in getting more compre-. hension than doing speed reading. The reader does not try to increase his speed merely because he worries about loosing comprehension. He thinks that by doing speed reading, the comprehension will decrease. That is why some readers can not improve his speed in reading.

By knowing some techniques of how to read faster and some points that hamper the speed reading, the reader can improve his rate of reading speed. This is sertainly adjusted with the various types of reading material.

2.4. Vocabulary Enrichment

Before talking about the vocabulary enrichment, at first the writer would like to put forward some definitions of vocabulary.

Burns and Schell (1969:959) defined the vocabulary as "the stock of words used by a per-

son, class, or profession ".

Hornby (1979:969) stated that " vocabulary is the total number of words which make up a language. Vocabulary is the range of words to, or used by a person in trade, profession, etc. ".

Based on the definition above, the writer concludes that vocabulary is the total number of words used by a person or group of people in order to build up a language. In relation to this, Heilman (1976:14) says that words serve as symbolic representatives of both object and idea. He also pointed out that what we call words would be nothing more than vocal noises except for the agreement that those noises have particular meanings. Therefore a word is more than just a combination of sounds or group of letters, but it is a symbol that enable men to communicate with each other.

Vocabulary by some experts is usually distinguished into two kinds, namely recognition vocabulary and active vocabulary, or sometimes called as 'working vocabulary'.

Brewton (1962:53-54) described the recognition vocabulary as " the total stock of words a person knows well to understand when he meets them in context. He may not able to define all

son, class, or profession ".

Hornby (1979:969) stated that " vocabulary is the total number of words which make up a language. Vocabulary is the range of words to, or used by a person in trade, profession, etc. ".

Based on the definition above, the writer concludes that vocabulary is the total number of words used by a person or group of people in order to build up a language. In relation to this, Heilman (1976:14) says that words serve as symbolic representatives of both object and idea. He also pointed out that what we call words would be nothing more than vocal noises except for the agreement that those noises have particular meanings. Therefore a word is more than just a combination of sounds or group of letters, but it is a symbol that enable men to communicate with each other.

Vocabulary by some experts is usually distinguished into two kinds, namely recognition vocabulary and active vocabulary, or sometimes called as 'working vocabulary'.

Brewton (1962:53-54) described the recognition vocabulary as " the total stock of words a person knows well to understand when he meets them in context. He may not able to define all

the words and perhaps many of them are never used in speaking and writing. But, however, when he can give correct interpretation to the words ace cording to their context, they belong to their recognition vocabulary. Active vocabulary (working vocabulary), on the other hand, is the stock of words that a person actually uses in speaking and writing. Therefore, this kind of vocabulary is much less than the recognition vocabulary ".

bulary is the vocabulary enrichment by acquiring new words from the reading materials. This is because in reading a context, students face many new and, sometimes, difficult words. They should work out a way of handling them. They should try to figure out the meaning of many words from the way they are used in sentences. And as they have met new words repeatedly in a variety of context settings, hope they have enrich their vocabulary or their store of words.

There are various ways of enriching vocabulary which people or students can work with them.
One of them, as stated by Thomas and Robinson
(1982:11) is by direct and indirect experiences.
They gave example such as a field trip for junior
high school students to an airport can give mean-

the words and perhaps many of them are never used in speaking and writing. But, however, when he can give correct interpretation to the words ace cording to their context, they belong to their recognition vocabulary. Active vocabulary (working vocabulary), on the other hand, is the stock of words that a person actually uses in speaking and writing. Therefore, this kind of vocabulary is much less than the recognition vocabulary ".

What the writer means by the extent of vocabulary is the vocabulary enrichment by acquiring new words from the reading materials. This is because in reading a context, students face many new and, sometimes, difficult words. They should work out a way of handling them. They should try to figure out the meaning of many words from the way they are used in sentences. And as they have met new words repeatedly in a variety of context settings, hope they have enrich their vocabulary or their store of words.

There are various ways of enriching vocabulary which people or students can work with them.
One of them, as stated by Thomas and Robinson
(1982:11) is by direct and indirect experiences.
They gave example such as a field trip for junior
high school students to an airport can give mean-

ing to the words beacon, ramp, dispatcher, and meteorologist. The visit of social studies class to a settlement house can give depth of meaning to the words deprived area, immigrant, tenement, slum clearance, and social reform is another example.

Vocabulary, according to another experts, is also can be built up by the variety of our interest and participating in many activities. For exsample we recognize the words like curriculum, study period, gym, prom, and debate because of our interest in school activities. People recognize the words like carburetor, piston, power steering, and ignition because of their interest in automobiles.

In addition, Brewton and his friends (1962: 56) present two certain specific techniques that can be practiced and will enable us to grasp the meaning of unfamiliar words. The first of these technique is the use of context clues and the second is the word analysis.

2.5.1. Context Clues

A context clue is a hint to the meaning of a word given by its setting. Or in other word, using context clue in word recognition means figuring out the meaning of the word based on clues in the

surrounding context. These clues could be found in the sentence with the word, in the sentence before, or in the sentence following the word.

Context clues will be easy to be recognize by the readers if they are familiar with some of the kinds of clues often found in sentences.

The following are some common kinds of clues and explanation of each as presented by Kustaryo (1988:28):

a. Restatement Clue

A restatement clue is merely stating the word in another way, and usually the terms are very simple. It is usually set of by commas.

For examples:

- The <u>perimeter</u>, or the distance around the circle was exactly sixteen inches.
- Idolatry, the worship of idols, was practiced by many primitive people.

Perimeter, in the first example, is the distance around the circle. Idolatry, in the second example, means the worship of idols.

b. Synonyms Clue

A synonym is one of two or more words that have the same meaning. When an author uses a difficult word, he often use a familiar word to make it easy to understand. For instance;

- The old man was <u>cantankerous</u>. He was ill-tempered, meam, and extremely quarrelsome.
- At first I was doubtful that I could do the job. After one successful week, however, I am much less <u>dubious</u>.

In the first example the word cantankerous means ill-tempered, mean, and quarrelsome. This example is also an instance where the unfamiliar word appear early in the sentence. Therefore the reader must continue to the end of the sentence to realize the meaning of the difficult word.

The second example uses the word dubious which means doubtful. And this instance the unfamiliar word appears at the end of the sentence. Therefore, the reader must go back to the beginning of the sentence to realize the meaning of the difficult word.

c. Antonyms Clue

An antonym is a word of opposite meaning.

It can be identified by looking at the contrary words such as; but, in contrast to, etc.

For examples;

- The sea lion is a <u>cumbersome</u> animal on land, but in the water it is one of the most graceful. In contrast to their enjoyment of their country home, the Jones family found it difficult to adjust to <u>urban</u> life.

In the example above, the contrary word but of the first example explain that an opposite is being used. So the word cumbersome is the opposite of graceful which is means 'clumsy'. In the second example the contrary word. 'incontrast to' is used in order to indicate that the word urban is the opposite of the word country home.

d. Definition Clue

It is usual for an author to put in a ...
statement that clearly defines a difficult word.

In the following example the word incoherent
means not logically connected. While the word
contagious means spreading from one person to
another person. Those examples are;

- An <u>incoherent</u> statement is a statement that is not logically connected.
- A disease that can be spread from one person to another is said to be contagious.

e. Explanation Clue

Difficult words are sometimes explained to make the meaning clearer. The explanation migh be found in the same, before, or in the sentence following the word. For examples;

- The couple decided to buy the old house because their meager savings were short of the price of a new house.
- Marilyn is a <u>versatile</u> musician. Not only does she sing and play the piano, but she plays many other instruments as well.

In the first example above, the word meager is explained in the same sentence and is
signaled by the word because. In the sentence
the word meager means small or inadequate. In
this example the reader may have to use information both before and after the appearance of
the unfamiliar or difficult word to realize its
meaning.

In the second example the word versatile
is explained by additional information that
Marilyn is competent in many aspects of music.
f. Relationship Clue

This type of clue is less concrete and requires more thinking on the part of the learner. The reader must see the a relationship between the difficult word and somethings the author has stated. This relationship must be seen with little explanation and few clues. For examples; - The students were jubilant when they learned that their school had placed fist in the com-

petition.

- Bach time you drive in excess of the speed limit, especially when traffic is heavy, you <u>leopardise</u> not only your own life but the lives of other as well.

In the first example we have to think about how students would feel if they placed first in a competition. We have to find some synonyms for jubilant. They would be joyful, rejycing, or elated.

In the second example, likewise, the reader must see the relationship between excessive speed and a danger to his life. So, the word jeopardise means endanger.

.2.5.2. Word Analysis

Word analysis means dividing an unfamiliar or difficult word into its parts in order to arrive at a clear idea of what the word means. Generally words divided into two main parts, namely; root and affixes. Furthermore, the affixes consist of prefix and suffix. For more explanation about them is as follows:

a. Root

The root is the basic element of any word that give it its meaning. By understanding the root and the way they are used to build words,

one will be able to find the meaning of a large number of English Words. For example if a reader knows that the root 'audio' means 'hear' or related to the sound', then he will know that the word 'audiolingual' must have something to do with ear. In this case the word audiolingual means to hear the sound or it is relating to or involving listening and speaking in learning a language.

The purpose of introducing the concept of root is to give the reader building blocks to develop reading strategies.

b. Prefix

A prefix is one or more syllable that is placed before the root to form a new word. The connecting of the root to the prefix changes the meaning of the word slightly. Some prefixes have more than one meaning. Therefore a reader should always consider the context of the word when determining the meaning. For example take the prefix contra which may be means against and opposite.

c. Suffix

A suffix is a syllable or syllables added to the end of a root in English. This addition also changes the meaning of the root slightly. Neufeld and Webb in Koestaryo (1988:66) said that " as well as slightly changing the meaning of the root, the suffix plays a very important role in reading; it tells the function of the word, that is the way the word is used in the sentence ".

Beside the suffix can change the meaning of the word, it also changes the part of speech of the word. For example one meaning of the word 'to breathe' is to inhale or exhale air and is a verb. And by adding the suffix 'er' to the 'breathe' modifies the meaning to one that breathes, and results in the word 'breather' which is a noun.

simply using the dictionary. This is supported by Koestaryo (1988:3) the one who said that in order to minimize the students problems, they should be obliged and trained to use both English-English Dictionary and bilingual dictionary (English-Indonesian or Indonesian-English Dictionary) which are really useful for them to maintain either the meaning of the words reading context or type of the sentence in which the words are constructed. The result of this activity will strenghten the students' willingness to learn not only the English, but other discipline as well.

Neufeld and Webb in Koestaryo (1988:66) said that " as well as slightly changing the meaning of the root, the suffix plays a very important role in reading; it tells the function of the word, that is the way the word is used in the sentence ".

Beside the suffix can change the meaning of the word, it also changes the part of speech of the word. For example one meaning of the word 'to breathe' is to inhale or exhale air and is a verb. And by adding the suffix 'er' to the 'breathe' modifies the meaning to one that breathes, and results in the word 'breather' which is a noun.

simply using the dictionary. This is supported by Koestaryo (1988:3) the one who said that in order to minimize the students problems, they should be obliged and trained to use both English-English Dictionary and bilingual dictionary (English-Indonesian or Indonesian-English Dictionary) which are really useful for them to maintain either the meaning of the words reading context or type of the sentence in which the words are constructed. The result of this activity will strenghten the students' willingness to learn not only the English, but other discipline as well.

The most important thing is to know how the words are used in context or sentence for the meaning of a word changes in different context.

Take for example the word 'go' that gives various different indonesian equivalence in the following utterances:

"English : Give me a go at the job.
Indonesia : Berikan saya <u>kesempatan</u> untuk
melakukan pekerjaan itu.

English : It is his go. Indonesia : Itu gilirannya.

English : I am to go.

Indonesia : Saya yang akan pergi.

English : That country must go. Indonesia : Negeri itu pasti hancur.

English : My hearing is going.

Indonesia: Pendengaran saya mulai hilang.

English : This road goes to Elmira. Indonesia : Jalan ini menuju Elmira.

English : How does the poem go ?

Indonesia: Bagaimana <u>bunyi</u> sajak itu ?

English : To go by car. Indonesia : Naik mobil

English : Everything goes. Indonesia : Apapun boleh.

English : The gold medal goes to you.

Indonesia: Medali emas itu jatuh ketangan-

English : Ten employees will have to go.
Indonesia : Sepuluh orang pegawai harus diberhentikan.

English : Where does this box go ? Indonesia : Dimana tempat kotak ini ?

English : How did the tennis go ?

Indonesia : Bagaimana permainan tennis itu ?

English : A fuse went.

Indonesia : Sekering putus, etc. ".

Echols and Shadily (1975:272).

Those examples above show us how the word go is varied according to the context or sentences where the word appears. It proved that we have to be aware in learning vocabulary, or words, in isolation bacause it is not helpful as it is expected. Or, in other word, we can say that by building up vocabulary through reading, the students will understand how each word is used in context. Thus they will be prevented from misusing it.

Although it is advisable to discourage students from making use of dictionary, but the usual tendency is to use it too often. And as a matter of fact, some students spend much time to find a word in a dictionary. Furthermore, if a student has to use a dictionary, he had better be able to use it so efficiently.

The first step towards using a dictionary as a tool is deciding which word must be look up. Having concluded that a word must be looked up, we have to make the best use of the information the dictionary supplies as quickly as possible.

To better understand the use of the dictionary, it is necessary to learn some of the common terms and their meaning that is often found in a dictionary. Those common terms are as follow:

a. Entry Words

a. Entry Words

Entry words are the terms in the dictionary for which a definition and explanation are given. Entry words are arranged in alphabetical order innthe dictionary and are usually in boldface type. For examples; accolade, accomodate, accomodating.

b. Guide Words

Guide words are usually put at the top of the page to guide the user in finding a word alphabetically. The guide word in the left is the first entry word on the page, and the one on the right is the last entry word.

c. Syllabification

Syllabification is shown in the entry word and every syllable are separated in the word by a dot. For instance; ac-co-lade, ac-com-mo-date, ac-com-mo-dat-ing.

d. Pronunciation

The phonetic pronunciation, are given inside parentheses following the word. The key to pronounce well is put in the parentheses to guide a user in pronouncing unfamiliar words. For examples are as follows; accolade ['ækəleid] accommodate [ə'kɒmədeit]

accommodating [a kpmadeitig] But in the present dictionary, the parentheses replaced by slanting line.

e. Parts of Speech

The parts of speech of the entry word is indicated immediately following the pronunciation of the word. For instance; accolade / aekaleid / n. (noun) accommodate / a kpmadeit / v. (verb) accommodating / a'komadeitin / adj. (adjective)

f. Illustration

Picture of lesser-known objects, maps, and so fort, are given so that the readers get a clear concept of the definition.

g. Multiple Meanings

Most of the word in English have more than one meaning. When a reader look up the meaning of the new word, he must choose the meaning that fits the way the word is used in the sentence context. For instance look back to the word 'qo'.

2.5. The Correlation Between Reading Speed and The Extent of The Vocabulary.

In the previous paragraphs of this chapter, we have seen that reading speed and the extent of vocabulary are two indivisible variables. Reading is one of the doors to arrive to the richness of vocabulary. Thomas and Robinson (1982:12,30) stated out that wide reading increases vocabularies. That most of the vocabulary growth of capable readers is gained through their reading. They have met new words repeatedly in a variety of context setting, and gradually the words become a part of their vocabularies. And on the other hand, they also pointed out that students will be a better, faster reader with more word power.

Zorn (1980:109) said that reading speed and the knowledge of vocabulary go together like bread and butter. They cannot be separated. It proves that context is one of the major tools for students in vocabulary expansion. While the students who have more word power will be a better faster reader.

Based on the statement above, the writer concludes that speed reading should be followed by at least 50% of vocabulary enrichment. It is supported by Nurhadi (1987:42) who said that if a speed reader gets below 50 % of vocabulary enrichment, he is a bad speed reader. And if he gets higher, he categorized as a good speed reader.

To see the correlation between reading speed and the extent of the vocabulary, the writer puts forward the standard of reading speed of the whole

types of reading material as follows :

Table 2

The satndard of reading speed of the whole types of reading materials.

Aproach	Haw Fast	When to Use
Scanning, not a true reading rate, just glancing un- till you find what you want.	maybe 1500 or more words per minute (rate is an individual matter),	when glancing down pages to find a single of information.
Skimming, not a true reading rate, just getting the gist of the article, hitting the high points.	maybe 800-1000 words per minute	to get the general content of an art-icle, what is all about.
Actual Reading Rate 1. Very rapid	maybe 400 words or more per minute	for easy, fast mov- ing fiction (enter- taiment reading).
2. Rapid	maybe 300 words per minute	for fairly easy materials when you want only the more important ideas.
3. Average	perhaps 250 words per minute	for magazine art- icles, travel books novels, etc.
4. Slow and care- ful	from 250 words per minute-all the way down to a slow 50 words per minute or even slower.	for difficult con- cepts and vocabula- ry, for through read ing of technical material, for cont- aining every detail for weighing the truth of difficult reading.

Lamar and Robinson (1982:176).

The relationship between reading speed and the extent of vocabulary is easy to be understood. That is to say that if a reader is going to read an easy

reading material, he will read it in his maximum rate without loosing the massage of the material. It is becouse he knows almost all of the words occure in the material he read. But, on the other hand, if he finds some difficult vocabularies in the material he read, they will hamper him to come to his maximum rate of his speed reading.

The reading speed and the extent of the vocabulary are said to correlate positively if a reader can read certain type of reading material with a speed rate that conforms the list with at least 50 % of vocabulary enrichment. It means that the reading speed and the extent of the vocabulary corelate negatively if they are under the standard mentioned.

CHAPTER III

DATA ANALYSIS AND DISCUSSION

This chapter deals with data analysis of reading speed, vocabulary enlargement, and the questionnaire as well. The result of the research will be discussed in the last part of this chapter.

3.1. Presentation of Data Analysis

3.1.1. The Analysis of Data Collected Through Reading Speed Test.

The following table shows the result of the students' reading speed in words per minute.

The formula used in analyzing data is pre - sented at first.

Number of words
have been read

Number of time
used (second
elapsed)

Table 3

The raw result of the students' reading speed in words per minute

No	Reg. Numb.	Reading I One Night On The Mt. Everest Of Man- Made Structure (520 words)		Reading II Thanksgiving Day (600 words)	
-		Time	WPM *	Time	WPM *
1	2	3	4	5	6

1	2	3	4	5	6	
2	88 003	05,00	104	05,00	120	
3	012	05,00	104	13,00	46	
4.	016	05,00	104	10,00	60	
5	024	05,00	104	07,00	85	
6	025	05,00	104	03,00	200	
.7	027	05,00	104	08,00	75	
8	032	05,00	104	05,00	120	
9	034	08,00	65	06,00	100	
10	037	05,00	104	06,00	100	
11	041	05,00	104	05,00	120	
2	042	05,00	104	05,00	120	
.3	044	05,00	104	05,00	120	
4	045	05,00	104	05,00	120	
5	050	06,00	86	06,00	100	
6	051	05,00	104	04,00	150	
7	055	. 04,00	130	05,00	120	
8	065	05,00	104	10,00	60	
9	066	05,00	104	06,00	100	
0	071	05,00	104	02,00	300	
1	073	05,00	104	07,00	85	
2	075	05,00	104	06,00	100	
3	080	05,00	104	07,00	85	
4	083	07,00	74	08,00	74	
5	085	05,00	104	05,00	120	

1	2				
1			4	5	6
26	88 088	06,00	86	06,00	100
27	092	05,00	104	03,00	200
28	107	04,00	130	03,00	200
29	121	05,00	104	07,00	85
30	148	07,00	74	06,00	100

WPM * = word per minute

Based on the table above, we can conclude that from the two reading speed tests a few of the students can reach the reading speed between 150 to 200 words per minute. In reading one, there is no one of the students can reach until 150 words per minute. Two of them can reach 130 words per minute only.

Nevertheless, in reading two, 5 of them (16,66%) can reach between 150 to 200 words per minute. And the rest students are as follows:

All of the students, in reading one, can only reach between 65 to 130 words per minute, or 100 %. And in reading two, 25 (83,33 %) of the students reach 46 to 120 words per minute in their speed reading.

3.1.2. The Analysis of the Data Collected Through Vocabulary Enrichment

1 2						
		3	4	5	6	
26	88 088	06,00	86	06,00	100	
27	092	05,00	104	03,00	200	
28	107	04,00	130	03,00	200	
29	121	05,00	104	07,00	85	
30	148	07,00	74	06,00	100	

WPM * = word per minute

Based on the table above, we can conclude that from the two reading speed tests a few of the students can reach the reading speed between 150 to 200 words per minute. In reading one, there is no one of the students can reach until 150 words per minute. Two of them can reach 130 words per minute only. Nevertheless, in reading two, 5 of them (16,66%) can reach between 150 to 200 words per minute. And the rest students are as follows:

All of the students, in reading one, can only reach between 65 to 130 words per minute, or 100 %. And in reading two, 25 (83,33 %) of the students reach 46 to 120 words per minute in their speed reading.

3.1.2. The Analysis of the Data Collected Through Vocabulary Enrichment The following table presents the raw data of the result of the students' vocabulary enrichment by using the formula of percentage as follow:

Sum of correct numbers

Total number of question's items

X 100 =VE*=%.

Every student has two data of vocabulary enrichment as in reading speed data.

Table .4

The raw data of the students'

vocabulary enrichment.

No.	. Reg.	.dme	Vocabulary Enrichment of Everest in %	E Mt.	Vocabulary Enrichment of Thanksgiving Day in %
1	2		3		. 4
1.	88 0	01	73		60
2	88 0	03	93	-	80,
3	88 0	12	73		60
4	88 0	16	80		60
5	88 02	24	80		53
6 .	88 03	25	80		53
7	88 02	27	53		53
8	88 03	32	80		47
9	88 03	34	67		73
10	88 03	37	во		73

escondulus.					
1		2	3	 	
11	88	041	73	4	
12	88	042	. 73	47	
13	88	044		73	
14		045	53 53	60	
15		050		60	
			67	47	
16	88	051	67	67	
17	88	055	47	33	
18	88	065	87	67	8
19	88	066	80	67	10
20	88	071	40	33	*
21	. 88	073	53	53	
22	88	075	53	40	
23	88	080	во	67	
24	88	083	80	60	
25	88	085	67	47	
26	88	088	60	60	2 %
27	88	092	33	40	
28	88	107	80	60	
29	88	121	80	60	
30	88	148	80	60	

The table above shows that in the first vocabulary enrichment test, 27 of the students (90 %) are able to reach between 50 to 93% of vocabulary enrichment. There are three of them (10%) can reach only between 33 to
47 % of their vocabulary enrichment. While
in the second vocabulary enrichment test,
22 of the students (73,33%) can reach
between 53 to 80 % of vocabulary enrichment.
There are only 8 of them (26,67%) able to
reach between 33 to 47 % of their vocabulary enrichment.

In order to determine the average ability of the students in reading speed and
vocabulary enrichment, the writer firstly
analyses both the data from reading speed
test and the data from vocabulary enrichment
test. In doing this, the writer uses the
following formula:

Where a and b are the scores of every reading speed and vocabulary enrichment tests from each student.

The result of this is shown in the follows ing table.

Table 5

The students average value of reading speed and vocabulary enrichment

No.	Registration	Reading Speed in WPM	Vocabulary en- richment in %	
1	Number 88 001	112	66,5	

1	2	3	A EMPUSTMENT 4		
2	88 003		36,5		
3	88 012	75	40		
4	88 016	82	57		
5	88 024	94,5	66,5		
6	88 025	152	66,5		
7	88 027	89,5	77		
8	88 032	112	63,5		
9	88 034	82,5	56,5		
10	88 037	102	76,5		
11	88 041	112	60		
12	88 042	112	73		
13	88 044	112	70		
14	88 045	112	70. ~		
15	88 050	93	57		
16	88 051	127	47		
17	88 055	125	66,5		
18	88 065	82	. 53		
19	88 066	102	73,5		
20	88 071	202	86,5		
21	88 073	94,5	. 53		
22	88 075	102	60.		
23	88 080	94,5	73,5		
24	88 083	. 74	36,5		
25	88 085	112	70		
26	88 088	93	46,5		

	-	3	4
27	88 092	152	70
28	88 107	. 165	70
29	88 121	94,5	70
30	88 148	87	56,5

In the above data, we can see that out of 30 students, only 5 of them (16,67%) are catagorized as good readers because they can reach between 120 to 150 words per minute with at least 50 % of vocabulary enrichment. That is if the category taken for the students untrained. Nevertheless, if the category is taken after those students trained, the result of the table will say to as that out of 30 students, only 4 of them (13, 33 %) are categorized as good readers because they can reach between 150 to 200 words per minute with at least 50 % of vocabulary enrichment. Thus, this category is based on the opinions of reading experts as have been discussed in chapter two.

To know the general capability of the sixth semester students of the Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang in reading speed and vocabulary
enrichment, the writer analyzed the data of reading
speed and vocabulary enrichment by using the Mean score
procedure. The result of this is shown in the following
tables:

Table 6
The mean score of reading speed

Reading Speed Score	Frequency	Deviation	FD
200 - 209	1	9	9
190 - 199	0	8	0
180 - 189	0	7	. 0
170 - 179	0	6	0
160 - 169	. 1	5	5
150 - 159	2	4	8
140 - 149	0	3	0
130 - 139	0	2	0
120 - 129	2	1	2
110 - 119	8	0	0
100 - 109	3	-1	- 3
90 - 99	6	-2	-12
80 - 89	5	-3	-15
70 - 79	2	-4	-8
Total	30	_	-14

These data are analyzed by using the formula of mean score as follow:

$$M = WM + (\frac{\sum fd}{N}) i$$
The result is
$$= 114.5 + (\frac{-14}{30}) 10$$

$$= 109.83$$

$$= 110 WPM$$

Table 7
The mean score of vocabulary enrichment

vocabulary en-			
richment score	F	D	FD .
80 - 89	1	1	1
70 - 79	11	0	0
60 - 69	8	-1	-8
50 - 59	6	-2	-12
40 - 49	2	-3	-6
30 - 39	2	-4	-8
total	N=30		-33

The result of the table is as follow :

$$M = WM + (\frac{\sum FD}{N}) i$$

$$= 74.5 + (\frac{-33}{30}) 10$$

$$= 63.50 \%$$

Those letters above stand for; M= mean

WM= work mean

D= deviation from WM

F= frequency

N= number of sample

Based on the tables above (table 6-7) we can see that the average of the students' reading speed and vo-cabulary enrichment are 110 WPM and 63.50 % vocabulary enrichment in general.

In order to make the writer easier to analyze these data in Pearson's product moment correlation, the data are ordered based on the reading speed in word per minute. The result is in the following table.

Table 8

Data presentation of the correlation product moment analysis

No.	Reg. Numb.	Reading speed in word per minute (X)	Vocabulary enrichment in percent (Y)	Statement
1	2	3	4	5
1	88071	202	86.5	boog
2	88107	165	70 -	good
3	88025	152	66.5	good
4	88092	152	70 -	good
5	88051	127	4.7	bad
6	88055	125	66.5	good
7	88001	112	66.5 4	bad
8	88003	112	36.5	bad
9	88032	112	63.5	bad
10	88041	112	60 -	bad
11	88042	112	73 ^	bad
12 .	88044	112	70 14	bad
13	88045	112	70 ~	bad
14	88085	112	70 :	bad
15	88037	102	76.5	bad
16	88066	102	73.5	bad
.7	88075	102	60	bad
.8	88024	94.5	66.5	bad
9	88073	94.5	53	bad
0	88080	94.5	73.5	bad
1	88121	94.5	70 %	bad

1	2	2		
	A AND THE WAY AND THE STATE OF	3	4	5
22	88050	93	57	bad
23	88088	93	46.5	bad
24	88027	89.5	77	bad
25	88148	87	56.5	bad
26	88034	82.5	56.5	bad
27	88016	82	57	bed
28	88065	82	53	bad
29	88012	75	40 .	bad
30	88083	74	36.5	bad

In the table above, the writer has classified the good readers and the bad readers. A reader is considered to be a good reader if he is able to read at least 120 words per minute without ignoring at least 50 % of vocabulary enrichment. This is the target of the untrained students. And after training, a good reader must be able to read at least 200 words per minute without ignoring at least 50 % of vocabulary enrichment. This is supported by Nuttle and Smith (1988:45) who say that the average reading speed for EFL students is 120 to 150 words per minute before training, and after training it is hoped that they could read at about 200 words per minute. And the students of the Pondok Pesantren are untrained students.

Based on Nuttle and Smith's statement and the stu-

dents condition, the writer classified as a good speed reader if he able to read between 120 to 200 words per minute with at least 50 % of vocabulary enrichment. It is supported by Nurhadi (1987) the one who say that speed reading should be followed by 50 % of vocabulary enrichment minimally.

Guided by the standard of good speed reader above, table 7 shows us that out of 30 of the students who were taking the test, only 5 of them (16,67%) are classified as good speed readers and the rest are bad.

3.1.3. The Analysis of Data Correlation

The following description presents the correlation between reading speed and the extent of vocabulary.

In analyzing the data, the researcher considered the reading speed as an independent variable (X) and the extent of vocabulary as the dependent variable (Y). The two variables are analyzed by using The Person's Product Moment Coefficient of Correlation Analysis. Before analyzing these data, they are tabulated first as in the following table.

Table 9

Preparation table in looking for the correlation between reading speed and the extent of vocabulary

No.	Reg.	Y	×	×2	Y	Y	y ²	xy
-	No.							

Preparation table in looking for the correlation between reading speed and the extent of vocabulary

No	Reg. Numb.	x	×	×2	Y	У	y ²	ху
1	2	3	4	5				
1	88071	200			6 ,	7_	.8	9
			93.3	8704.89	86.5	23.87	7 569.7769	2227.071
2	88107	165	56.3	3169.69	70	7.37	54.3169	414.931
3	88025	152	43.3	1874.89	66.5	3.87	14.9769	
4	88092	152	43.3	1874.89		7.37		
5	88051	127			3,000			
	10000000000			334.89	57	-5.63	31.6969	-104.155
6	88055	125	16.3	265.69	66.5	3.87	14.9769	63.081
7	88001	112	3.3	10.89	66.5	3.87	14.9769	12.771
8	88003	112	3.3	10.89	36.5	-26.13	682.7769	-86.229
9	88032	112	3.3	10.89	63.5	0.87	0.7569	2.871
10	88041	112	3.3	10.89	60	-2.63	6.9169	-8.679
11	88042	112	3.3	10.89	73	10.37	107.5369	34.221
12	88044	112	3.3	10.89	70	7.37	54.3169	24.321
13	88045	112	3.3	10.89	70	7.37	54.3169	24.321
14	88085	112	3.3	10.89	70	7.37	54.3169	24.321
15	88037	102	-6.7	44.89	76.5	13.87	192.3769	-92,929
6	88066	102	-6.7	44.89	73.5	10.87	118.1569	-72.829
7	88075	102	-6.7	44.89	60	-2.63	6.9169	17.621
8	88024	94.5	-14.2	201.64	66,5	3.87	14.9769	-54.954
9	88073	94.5	-14.2	201.64	53	-9.63	92.7369	136.746
0	88060	94.5	-14.2	201.64	73.5	10.87	118.1569	-154.354
	88121	94.5	-14.2	201.64	70	7.37	54.3169	-104.654

1	2	3	4	5				
1					- 6	7	8	9
22	88050	93	-15.7	246.89	57	-5.63	31.6969	88.391
23	88088	93	-15.7	246.89	46.5	-16.13	-27-22-0 (27-27-2	253.241
24	88027	89.5	-19.2	368.64	77	14.37	5178 TS	-275.904
25	88148	87	-21.7	470.89	56.5	-6.13	37.5769	133.021
26	88034	82.5	-26.2	686.44	56.5	-6.13	37.5769	160.606
27	88016	82	-26.7	712.89	57	-5.63	31.6969	150.321
28	88065	82	-26.7	712.89	53	-9.63	92.7369	257.121
29	88012	75	-33.7	1135.69	40	-22.63	512.1169	762.631
30	88083	74	-34.7	1204.09	36.5	-26.13	682.7769	906.711
	N= 30	3261	0	23037	1879	0	4206,467	5226.324

The table above shows that;

$$N = 30$$

 $X = 3261$ $Y = 1879$
 $\overline{X} = 108.7$ $\overline{Y} = 62.63$
 $\sum x^2 = 23037$ $\sum y^2 = 4206.467$ $\sum xy = 5226.324$

Based on these scores, we can find the correlation of the two variables by using the formula of coefficient correlation product moment as follows;

$$r_{xy} = \frac{\sum xy}{\sqrt{(\sum x^2)(\sum y^2)}}$$

$$r_{xy} = \frac{5226,324}{\sqrt{(23037)(4206,467)}}$$

$$r_{xy} = \frac{5226,324}{9844,002} = 0,530$$

$$r_{xy} = 0,530$$

After analyzing the data, it is found that the coefficient of correlation value (r_{xy}) is 0,530. To investigate wether this value is significant or not, the writer consulted with the table of critical value of (r) product moment on the basic of the 95% level of reliability or the 5% level of significance with a sample (N) = 30. Then it is found that the critical value of (r) with (N) = 30, and on the 95% interval of reliability is 0,463.

The investigation shows that the coefficient correlation of the two variables ($r_{\rm xy}$) is greater than the critical value of (r), or sometimes is written like this; $r_{\rm xy}$ critical value of (r), where ($r_{\rm xy}$) is 0,530 while the critical value of (r) with N= 30 on 95% or 5% level of significance is 0,463. It means that there is a significant correlation between reading speed and the extent of vocabulary of the third year students of Pondok Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang.

3.1.4. The Analysis Of Data Collected Through The Questionnaire

Till now we have seen that reading speed and the extent of vocabulary have significant correlation, where $r_{xy} >$ critical value of r.

Nevertheless, it does not mean that the third year students of the Pondok have high reading speed and vocabulary enrichment. Only a few of them have adequate reading speed and vocabulary enrichment as can be seen in the previous table of reading speed and vocabulary enrichment.

To know the supporting factors about the students' achievement in this research, the result of the questionnaire will inform us more.

The questionnaire consists 15 items, namely 13 items are close ended and the other, items number 14 and 15, are open ended. The data obtain from the thirteen close ended items will be tabulated at first, while the data obtained from the two other open ended items will be analyzed individually.

Table 10

1. Are you interested in learning English ?

Answer	Frequency	Percentage
a. very interested	. 22	73.33 %
b. interested	8	26.67 %
c. less interested	. 0	0,00 %
d. uninterested	0	0.00 %
total	- 30	100.00 %

The above table shows that from 30 students; there

are 22 students (73,33%) are very interested in learning English subjects; 8 of them (26,67%) are interested; and none of them are less interested or uninterested in learning the English subject.

Table 11

2. Do you often or ever found information about speed reading and the extent of the vocabulary?

Answer	Frequency	Percentage
a. often	11	36,67 %
b. sometimes	9	30,00 %
c. seldom	9	30.00 %
d. never	1	3.33 %
total .	30	100.00 %

The table shows that from 30 students; 36,67% (11 out of 30) of the students answered that they often find the information of reading speed and the extent of vocabulary; 9 of them (30,00%) answered sometimes; 9 of them (30,00%) of them answered seldom; and there is one out of the 30 students (3,33%) says that he never find or got information about the two.

Table 12

3. What is your opinion about the reading speed and the extent of vocabulary ?

7,000	1	Answer				Frequency	Percentage
a.	very	important	to	bе	learned	19	33,34 %

b. important to be learned	. 10	22 22 */
c. rather important	. 10	33.33 %
c. rather important to be learned	. 1	3.33 %
total	0	0.00 %
	30	100.00 %

The table above indicates that 19 out of 30 of the students (63,34%) said that reading speed and the extent of the vocabulary are very important to be learned; 10 of them (33.33%) said that it is important to learn the two subjects; one of them (3.33%) said that it is rather important and none of them (0.00%) stated that it is unimportant to learn the reading speed and the extent of vocabulary.

Table 13

4. Are you interested in improving your reading speed and enriching your vocabulary as well ?

Answer	Frequency	Percentage
a. very interested	18	60.00 %
b. interested	12	40,00 %
c. less interested	0	0.00 %
d. uniterested	0	0.00 %
total	30	100.00 %

The table above shows that from 30 students; 18 of them (60.00%) are very interested in improving their reading speed and enriching their vocabulary as well; 12 of them (40.00%) are interested; and none of them is less interested or uninterested in reading speed and vocabulary enriching.

Table 14

5. Do you believe that reading speed can be improved and vocabulary can be enriched ?

Answer		
YOSWEL	Frequency	Percentage
a. Yes, I do	30	100.00 %
b. No, I don't	0	0.00 %
total	30	100.00 %

From the data table above it can be seen that all of the students (100.00%) believe that reading speed can be improved and the vocabulary can be enlarged or enriched.

Table 15

6. Do you often read an English book, newspaper, magazine, or novel with a purpose to improve your reading speed and enrich your vocabulary?

Answer	Frequency	Percentage
a. often	6	20.00 %
b. sometimes	21	70.00 %
. seldom	3	10.00 %
l. never	0	0.00 %
total	30	100.00 %

This table shows that from 30 students there are; 6 of them (20.00%) often read English texts with a

purpose of improving both their reading speed and vocabulary; 21 of them (70.00%) sometimes do; 3 of them (10.00 %) seldom do; and none of them never do.

Table 16

7. What is the average length of time do you spend for reading in everyday ?

Answer	Frequency	Percentage
s. three hours or more	7	23.33 %
b. two hours	12	40.00 %
c. one hour	9	30.00 %
half hour or less	2	6.67 %
total	30	100.00 %

From the data table above it can be seen that 7 out of 30 students (23.33%) spend three hours of their time for reading an English reading material per day; 12 out 30 students (40.00%) spend two hours; 9 out of 30 students (30.00%) spend one hour; and there are 2 out of the 30 students (6.67%) who spend half hour or even less to read English material per day.

Table 17

8. How often do you go to the library for reading in a week ?

Answer	E.	Frequency	Percentage
three times Of	more	15	60.00 %

		30	100.00	%
\equiv	total	0	0.00	%
đ.	never	5	16.67	%
c.	once	10	33.33	%
b.	twice			

The above table shows that out of 30 students; 60.00% (15 out of 30) of them come to the library three times in a week; 33.33 % (10 out of 30) of them come twice in a week; 16,67% (5 out of 30) of them come once in a week; and, ofcourse none of them never come to the library for reading in a week.

Table 18

9. Before starting to read, do you often realize your reading purpose ?

Answer	Frequency	Percentage
a. often	22	73.33 %
b. sometimes	5	16.67 %
c. seldom	2	6.67 %
d. never	1	3.33 %
total	30	100.CO %

The table above said that out of 30 students;
73.33 % (22 out of 30) students answered often; 5,67%
(5 out of 30) students answered sometimes; 6.67% (2 out of 30) of the students answered seldom; and 3.33%
(1 out of 30) of the students answered never realize his purpose in reading.

10. Do you often do regression in your reading ?

Answer					
			Percentage		
a. often		_			
b. sometimes		5	16.67 %		
o. Dome cimes		18	60.00 %		
c. seldom	1.6	2.80			
		2	6.66 %		
d. never		5	16.67 %		
total		30	100.00 %		

This table informs us that from the 30 students; 5 of them (16.67 %) often do reagression in their reading; 18 of them (60.00 %) sometimes do; 2 of them or 6.67 % of them seldom do; and there are five of them never do regression in their reading.

Table 20
11. When you are reading, do you know what you read ?

answer	frequency	percentage
a. Yes, I do	8	26.67 %
b. No, I don't	0	0.00 %
c. sometimes 'Yes'	22	73.33 %
total	30	100.00 %

Obviously the table above shows that from 30 students, there are 8 of them realize well that they are reading, or 26.67 % of them. None of them do not realize that he is reading, but there are 22 of them (73.33%) who sometimes do and sometimes not.

Table 21

12. Do you often practice your speed reading and enrich

your vocabulary well ?

Answer		
awer	Frequency	Percentage
a. often	18	
b. sometimes	10	60.00 %
	8	26.66 %
c. seldom	2	6.67 %
d. never	2	
total		6.67 %
	30	100.00 %

This table shows us that from 30 students, there are 18 of them (60.00 %) often practice their reading speed and enrich their vocabulary as well. There are 8 of them (26.00 %) sometimes do the two. There are 2 of them seldom do, and also there are 2 of them never do. Each of the last two frequency is 6.67 in percent.

Table 22

13. When your teacher expalins the lesson in English, do you understand what is he explaining ?

Answer	Frequency	. Percentage		
a. Yes, I do	8	, 26.67 %		
b. No, I don't	0	0.00 %		
c. sometimes 'Yes'	22	73.33 %		

The table above indicates that most of the students (73.33 %) sometimes understand the teacher when he/she

is explaining the lesson in English and sometimes not. Furthermore the table tells us that there are 8 out of 30 students (26.67 %) who are sure that they understand the teacher when he/she is explaining the lesson in English. None of them says that he does not understand his teacher when He/She understand is explaining the lesson in English.

The following data are the open ended tests which will be analyzed individually;

The 14th item of the questionnaire is about;

What is your effort to increase your reading speed and enrich your vocabulary as well ?.

From this question, the writer got saveral answer as tabulated in the following table;

Table 23

Answers	Frequency	percentage
1. wide reading	27	90.00 %
2. memorisation	18	60.00 %
 practicing (speaking, listening, and writing 	1) · 13	43.33 %
4. consult the dictionary	5	16.67 %
5. asking the teacher	2	6.67 %

The table above indicates that from 30 students, . there are 27 of them (90.00 %) said that they try to increase their reading speed and vocabulary through or

by reading widely. There are 18 of them (60.00 %) answer that they try to increase their reading speed and vocabulary trhough memorizing the new words. There are 13 of them (43.33 %) answer that they try to increase their reading speed and vocabulary through practicing. Here, practicing means speaking, listening, and writing. There are 5 of them (16.67 %) said that in order to increase their vocabulary or when ever they find a new difficult word in reading materials, they try to consult the dictionary, and there are only two of the students (6.67 %) ask the teacher to solve their reading speed and vocabulary problems.

The last item of the questionnaire is; What factors hamper your reading speed and vocabulary enrichment? To make the writer analyze the item easily, the answers of the students are also tabulated as follows;

Table 24

	answers	frequency	percentage		
1.	lack of English reading material such as; books, magazines, newspaper, etc.	12	40.00 %		
2.	not enough time/there are too many lessons	12	40.00 %		
3.	laziness	6	20.00 %		
	friend's negative response	3	10.00 %		
	forgetfulness	2	6.67 %		

The table shows that from 30 students, there are 12 of them (40.00 %) answer that the lack of English reading materials hampers them to do speed reading and vocabulary enriching as well. There are also 12 of them (40.00 %) who said that they do not have enough time to do the two because they have too many lessons to study in the Pondok. There are 6 of the students (20.00 %) answer that the laziness hampers them to do the two; 3 of them (10.00 %) said that their friends negative response hampers them to be speed reader and enlarge their vocabulary as well; and 2 of them (6.67 %) said that forgetfulness also hampers them to do the two.

After all the data have been analized, it seems that the ability of a reader to read rapidly is interested by his knowledge or store of words. According to the testing of significant correlation critical value of (r) product moment of this research, there is a significant correlation, positive correlation, between reading speed and the extent of the vocabulary of the third year students of the Pondok Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang. Nevertheless, it does not mean that they all can read speedily with good vocabulary enrichment. This matter will be elaborated more clearly in the next section, namely, in the discussion of the result of the data analysis.

3.2. Discussion of Data Analysis

3.2. Discussion of Data Analysis

The result of the data analysis shows that from 30 students taken as the sample of this research, there are 20.00 % (6 out of 30) of them able to read rapidly; that is between 120 to 200 words per minute. And there are 63,33 % (19 out of 30) students who have good vocabulary enrichment; that is between 50 to 100 %. There are only 16,67 % (5 out of 30) students have good read - ing speed and vocabulary enrichment. They can read between 120 to 200 words per minute with 50 to 86 percent of vocabulary enrichment.

This indicates that generally the third year (sixth semester) students of the Pondok Pesan - tren IMMIM ujung Pandang have low reading speed, although almost most of them have a good vocabu - lary enrichment. The data analysis shows that there are only 16,67 % (5 out of 30) students who can read rapidly and enlarge their vocabulary as well. It means that 83.33 % (25 out of 30) students could not read speedily with a good vocabulary enrichment. It proves that almost all of the students read the reading material word by word in order not to loose the meaning of the words in the texts.

Having seen the fact above, the writer con - cludes that only a few (16.67%) of the third

year students of the Pondok Pesantren can be regarded as the good readers, or good speed readers. This is because there are still a lot of students who think that reading speed is not very important to learn (see table 12). The table indicates that there are 36.67 % (11 out of 30) students who said that it is important, or even rather important to learn how to read rapidly and enlarge the vocabulary better. Another 63.33 % (19 out of 30) students stated that it is very important to learn the two.

In the table 14, we can see that all of the students (100%) believe that reading speed can be improved and vocabulary can be enlarged. Nevertheless, (see table 13) there are only 60.00 % (18 out of 30) students who said that they are interested in improving their reading speed and vocabulary as well, while another 40.00 % (12 out of 30) students stated that they are not very interested in improving the two. That is why in table 15 most of the students (70.00 %) only sometimes, not often, read English books, newspapers, magazines or any other English materials. And in fact, there are 3 out of 30 students (10.00 %) who seldom read English materials. It means that only 20.00 % students (6 out of 30) often read Eng-



lish reading msterials.

In the table 17 we can see that there are 50, 00 % of the students often or always go to the library to read English texts; while in the table 16 only 23.33 % of them always spend three hours or more for reading everyday. In the table 19 there are 60,00 % of them sometimes do regression while they are reading; while in the table 20 there are only 26.67 % of them know well what they are reading. In the table 22 there are 60.00 % of the students often or always practice how to read rapidly and to enlarge their vocabulary; while in table 11 we can see that there are 46.67 % of the students who never find any information about the two. Obviously, it is logical that there are only 16.67 % (5 out of 30) students who can read between 120 to 200 words per minute with 50 to 85 % of vocabulary.

As in the table of interpretation of correlation, the result of the research is between 0,400 to 0.600. It means that the correlation between the two variables (reading speed and the extent of the vocabulary of the third year students of the Pondok Pesantren) is moderate because the result of the correlation analysis is 0,530. In this respect, it can be concluded that the reading speed and the ex-

tent of vocabulary of the students of the Pondok correlate positively. It proves that reading speed and vocabulary enrichment can not be separated. Thomas and Robinson (1982:30) state that the students will be a better, faster reader with more word power, while on the other hand wide reading increases vocabularies.

In the table 5 (the students average value of reading speed and vocabulary enrichment) we can see that there are 6 (20.00 %) out of 30 students who can read between 120 to 200 words perminute, however, not all of them can reach at least 50 % of vocabulary enrichment. On the contrary, the table also shows that there are 83.33 % (25 out of 30) students got 50 to 85 % vocabulary enrichment, but not all of them can read faster. It proves that most of the students are afraid to read rapidly because they still think that reading such a way will reduce the vocabulary enrichment.

There is one thing that is important to know that in this reasearch, the researcher found out that most of the students still have low reading speed, while most of them can enrich their vocabulary well. It is because they have been trained to enlarge their vocabulary for three years, while to

read speedly, they never been thought, eventhough some of them found information about how to read speedily.

CHAPTER IV

CONCLUSION AND SUGGESTION

4.1. Conclusion

Having analyzed the data in the previous chapters, the writer draws the following conclusions:

- a. There is a sufficient positive correlation between reading speed and the extent of vocabulary of the third year (sixth semester) students of the Pondok Pesantren Modern Pendidikan Al-Quran IMMIM Ujung Pandang.
- b. The degree of correlation (rxy) between reading speed and the extent of vocabulary is 0.530. And by consulting the table of critical value of (r) based on 95 % of the reliability or the 5 % level of significance and the population (N) 30, it is found that the critical value of (r) is 0,463. It means that the two variables have a positive significant correlation. The position of their significance is between 0.400 to 0.600 (moderate or rather low).
- c. The result of the tests show that the scores of the students' reading speed and vocabulary enrichment are low. We have seen that out of 30 students, only 16,67 % (5 out of 30) of them are able to read between 120 to 200 words perminute with at least 50 % of vocabulary enrich-

ment. And eventhough most of the students have a good vocabulary enrichment (see table 5), only six of them have a good reading speed. This is because most of them still think that speed reading can hamper a reader to enlarge the vocabulary.

4.2. Suggestion

In relation to those problems, the writer would like to put forward the following suggestions:

- a. The students of the Pondok need to be thaught reading speed. And the teacher should make a balance between the reading material and the studnets level in doing this subject.
- b. Speed reading and enriching vocabulary well should become one of the focuses or factors in teaching reading.
- c. As we can see in table 24, the Pondok is demanded to provide more English book or English reading materials in order to fulfill the students' need.

BIBLIOGRAPHY

- Prewton, E., John, et. al. 1962. <u>Using Good English</u>, volume 10, USA; Laidlow Brothers Publishers; a devision of Doubleday & Company, Inc.
- Echols, M., John and Shadily, Hassan, 1990. An English Indonesian Dictionary, 18th ed. and An Indonesian English Dictionary, 3rd ed. Jakarta; PT. Grammedia.
- Fry, Edward, 1967. Reading Faster, New York; The Combridge University Press.
- Hadi, Sutrisno, 1987. Methodology Research, Yogyakarta; Yayasan Penerbitan Fakultas Psikology, UGM.
- —) Hornby, A., S., 1974. Oxford Advanced Learner's Dictionary of Current English, 3rd ed., Great Britain; Oxford University Press.
 - Koestaryo, Sukirah, 1988. Reading Technique for college Students, Jakarta; P2LPTK
 - Lado, Robert, 1961. Language Testing; the construction and use of foreign language tests, 9th impression, Hong Kong; Wing Tai Cheung Printing Co. Ltd.
 - Maberly, Norman, C., 1966. Mastering Speed Reading. USA; American Library.
 - Nurhadi, 1987. Membaca Cepat dan Efektif, Malang; Sinar Baru.
- Panitia Penyusun Bahan Pelajaran Bahasa Inggris Untuk SLTA, Departemen Pendidikan dan Kebudayaan, 1979. English for the SLTA; student book II, Jakarta; Departemen Pendidikan dan Kebudayaan.
- Simanjuntak, Gloria, Edithia, 1988. Developing Reading Skills for EFL Students, Jakarta; P2LPTK.
 - Smith, Frank, 1987. Psycholingsistics and Reading, USA; Holt, Rineheart and Winston Inc.
 - Soedarso, 1987. Sistem Membaca Cepat dan Efektif, Jakats; PT. Gramedia.
 - Tompubolon, Dp., 1986. Kemampuan Membaca, Technik Membaca Efektif dan Efisien, Bandung; Angkasa.
- Thomas, J., B., and Terry Page, G., 1977. International

- Dictionaire of Education, New York; Kogan Page London/Nichols Publishing.
- Thomas, Lamar, Ellen and Alan, Robinson, H. 1982. Improving Reading in Every Class, USA; Allen and Bacon Inc.
- Thompson, Lionel, 1986. Reading for Indonesian Students, Book 2 (Reading Skills), LP-UNHAS.
- Wainwright, B. A., Gordon, 1972. Rapid Reading Made Simple, Britain; W.H. Allem and Co. Ltd.
- Zorn, L., Robert, 1980. Speed Reading, USA; a devision of harder and row.

APPENDIX A

TABEL NILAI-NILAI r PRODUCT MOMENT

N	Tarai	Signif		Tarai	Signi	f	Tarat	Signif
	95%	95%	N	95%	95	- N	95%	
3	0,997	0,999	26			+-	-	95%
4	0,950		27	0,388				0,345
5	0,878		28	0,381		0.12022	3	0,330
8	0.0 *40.0 5.0 7	-,,,,,	29	0,374			0,244	0,317
6	0,811	0,917	30	0,367		70	0,235	0 206
7	0,754	0,874	30	0,361	0,463	75	0,233	88 75
8	0,707	0,834	31	0,355	0,456		0,220	0,296
9	0,666	0,798	32	0,349	0,449	. 1990	0,220	0,286
10	0,632	0,765	33	0,344	0,442	1000	0,213	0,278
3.2			34	0,339	0,436		0,20,	0,270
11	0,602	0,735	35	0,334	0,430	O.E.	0,202	0,263
12	0,576	0,708		35933	-,	100	0,195	0,256
13	0,553	0,684	36	0,329	0,424	125	0,176	0,230
14	0,532	0,661	37	0,325	0,418	150	0,159	0,210
15	0,514	0,641	38	0,320	0,413	175	0,148	0,194
			39	0,316	0,408	Section .	LIGHT.	
16	0,497	0,623	40	0,312	0,403	200	0,138	0,181
17	0,482	0,606				300	0,113	0,148
18	0,468	0,590	41	0,308	0,398	400	0,098	0,128
19	0,456	0,575	42	0,304	0,393	500	0,088	0,115
0	0,444	0,561	43	0,301	0,389	600	0,080	0,105
.		0.540	44	0,297	0,384	700	0,074	0.007
1	0,433	0,549	45	0,294	0,380	800	0,070	0,097
2	0,423	0,537	16	0,291	0,376	900		0,091
3	0,413	0,526	46			900	0,065	0,086
4	0,404	0,515	47	0,288	0,372	1000	0.063	0.001
5	0,396	0,505	48	0,284	0,368	1000	0,062	0,081
			49	0,281	50. V			
			50	0,279	0,361			

(Quoted from Sutrisno Hadi 1979:360)

APPENDIX B

DEPARTEMEN PENDIDIKAN DAN KEBUDAYAAN
FAKULTAS SASTRA / SASTRA INGGRIS UNIVERSITAS HASANUDDIN
KAMPUS TAMALANREA KM.10 UJUNG PANDANG

KETERANGAN

Berikut ini akan disajikan kepada anda dua macam bahan bacaan yang diambil dari buku milik Departemen Pendidikan dan Kebudayaan yang berjudul "ENGLISH FOR THE SLTA". Masing-masing materi bacaan berjudul ONE NIGHT ON THE MT. EVEREST OF MAN-MADE STRUCTURE ...dan THANKSGIVING DAY yang memiliki 500 sampai 600 kata.

Melalui kedua bacaan ini akan terlihat sejauh mana tingkat reading speed dan vocabulary extending yang
anda miliki saat ini. Oleh karena itu anda diminta un tuk membaca materi bacaan tersebut secepat mungkin, sesuai dengan kemampuan anda, dengan tidak mengurangi per
hatian anda terhadap vocabulary (kata-kata sulit)
yang terdapat didalam kedua bacaan itu.

PETUNJUK

- Terlebih dahulu tulislah nama, nomor stambuk, dan starting time pada tempat yang telah disediakan sebelum anda membaca materi yang diberikan.
- Setelah selesai membaca, tulislah finishing time atau waktu dimana anda selesai membaca materi bacaan yang diberikan pada tempat yang telah disediakan.

ONE NIGHT ON THE MT. EVEREST OF MAN-MADE STRUCTURES

The tallest building built by man is the Empire State building in New York City, its 102 stories reaching almost one fifth of a mile into the sky.

During office hours thousands of people work in its office space. Till midnight, tourists from all over the world ride its elevator toward the observation tower on the top of the world's tallest skycraper. The terrace of the observation tower is high enough to enable sight-seers to enjoy a view covering an area of three hundred and more square miles.

Sometimes migrating birds come down on the tower to rest for a while during the night. Since there are not glaring lights coming from the tower, birds have seldom been killed by flying into its walls. Sometimes millions of insects cover the glass of the windows and the walls. No space is left uncovered.

Edwin Way Teale, a natural historian, got permission - which given very seldom - to spend one night
on the top of this " Mt. Everest " of man-made buildings. He was to enjoy not only listening to and looking at the birds as were flying past the tower, but he
would also enjoy all the sights and sounds in the
night above, around and below him.

Teale reached the terrace a little before mid-

night when the last sightseers were leaving. After midnight he had the terrace all to himself.

A quarter-moon was hanging in the sky as if be.side the tower top, and thousands of stars were winking
all over and above the city. Below, there was such as
great number glaring lights to see that the moon and
stars look pale.

The street lamps of the city - 156,000 of them - stood out in line and curves. On all sides, Teale could see lighted windows, that made the sky look pale. There was so much to see that only a patient natural historian like him could observe so many details.

He could not see the birds clearly but his trained ears told him what birds were flying over and above him.

Around two o'clock the life in the city, at last was slowing down. Most of the lighted windows disappear ed. Fewer and fewer car light could be see. The street lamps darker, the stars were getting brighter. The buildings began to look like stones see through water. It was getting to dark to see them clearly.

Between three and four, there was an interruption. Out of nowhere, suddenly a guard came upon him. The guard was so supraised to see anybody on the terrace at such an early hour that he almost arrested Teale. Lucki ly the natural historian could produce the letter of

permission that he had recieved. Otherwise, for the rest of the night, the birds would have flow past the tower without his being able to observe them. It was then nearly four o'clock in the morning. This was night also for New York city. Only the taxis never sleep.

Before five o'clock, the sky was getting brighter and brighter above the eastern horizon. A new day was being borm.

Teale went downward in the elevator to go home that morning thinking of all those tired birds also going down to rest.

520 words

Name		=	 Starting	Time	:	•••••
Reg.	Numb.	:	 Finishing	; Time	:	

THANKSGIVING DAY

When the first Englishmen settled in New England in 1620, their first year was full of hardships. Many of them died becouse the winter was much colder than they were used to. Although they suffered greatly, they worked hard to build their house and plant a few crops. These people believed in the power of God and prayed that He would lead them safely through their hardships in the new country.

When they harvested the first crops they were greatly surpraised becouse they had not expected a very good harvest. Now they have plenty of food to eat and everybody was happy. They have enough food for the next winter. Becouse of that the governor proclaimed a day of thanksgiving to God for having led them safely through their difficult first year.

They sent four men to the woods to find animals for the thanksgiving feast. They returned with a large number of animals most of which were turkeys. The feast lasted almost a week. A large number of friendly natives were invited to the thanksgiving feast.

By this time the small group of Englishmen were beginning to think that they were used to the weather of the new country. But suddenly in May 1623 it stop raining. The crops were almost destroyed. The governor then appointed a day in the middle of July for fasting

and prayer. It seemed that their prayers were not in vain, since soon after they had fasted and prayed the weather changed, and their crops were saved. Again a day of thanksgiving was proclaimed and a feast was given. And again it was the turkeys that were killed for food. Since the first thanksgiving feast it has become a tradition among Americans to always have turkeys for their thanksgiving dinner.

President Washinton issued a proclamation in 1789, appointing Thursday November 26 as a day of general thanksgiving. However, for many years many states celebrated the day at different times. A woman whose name was Mrs. Hale wrote many articles in magazines and many letters to governors and presidents suggesting that a day should be officially appointed so that Americans everywhere in the country could celebrate thanksgiving on the same day. At first she was not successful. But she was not discouraged, and continued writing for a period of not less than twenty years.

At last her articles and letters had results,

President Lincoln issued the second presidential proclamation in 1864, appointing the fourth Thursday in

November as a day of Thanksgiving. Since then Americans
have celebrated Thanksgiving Day on the fourth Thursday
in November.

Thanksgiving Day is an important holiday in Ameri-

ca. On this day stores, government offices, and schools are closed. It is celebrated in churches and in the homes. In the church a special service is given, special hymns are sung, and the presidential proclamation is read. In the homes the families have turkey for their thanksgiving dinner.

England had its first experience of the American thanksgiving in the Second World War when in 1942 churches in Britain were used by American armed force for the thanksgiving service. The 'Stars and Stripes' was carried to Westminster Abbey, the famous church in London. During the ceremony the United States ambassador read the presidential proclamation. Members of American military forces were there, many of whom took part in singing the hymns. They certainly did not forget God for helping the first Englishmen who settled in New England. What they seemed to forget at that moment were German bombs that might fall on them at any time.

600 words

Name		:	 Starting Time	:	• • • • • • • • • • • • • • • • • • • •
Rea.	Numb.		 Finishing Time	:	

APPENDIX C

Vocabulary Extending Tests

PETUNJUK SOAL :

Setelah menyelesaikan masing-masing materi bacaan yang diberikan kepada saudara, maka berikut in anda diminta untuk menjawab vocabulary extending test dengan petunjuk-petunjuk sebagai berikut:

- Masing-masing vocabulary extending test mempunyai 15 nomor pertanyaan.
- Tulislah nama dan nomor stambuk anda pada tempat yang telah disediakan sebelum menjawab ke lima belas nomor pertanyaan-pertanyaan berikut ini.
- 3. Bacalah baik-baik setiap teks pertanyaan yang diberikan kemudian jawablah/artikanlah kata-kata yang digaris bawahi sesuai dengan conteks pertanyaan yang diberikan.

Atas segala perhatian, partisipasi dan sumbangsih gng anda berikan demi suksesnya penelitian ini, penulis tidak lupa mengucapkan terima kasih. Vocabulary Extending Test; the first reading

- 1. Being the tallest skycraper, the Empire State Building <u>stands out</u> from the other building in the area. <u>stands out</u> means:
- 2. People can reach its top, which is almost one fifth of a mile, by elevator.
 top means:
- From morning until <u>midnight</u> hundreds of them go up to the observation tower to enjoy the beautiful scenery.

midnight means :

 The <u>space</u> in the terrace of the observation tower is large enough for many sightseers to enjoy the scenery.

space means :

- 5. The scenery look more beautiful in the evening when the sky is bright, not <u>covered</u> by clouds. <u>covered</u> means:
- 6. Edwin Way Teale, a <u>well trained</u> natural historian, went up to the tower to study the birds. <u>well trained</u> means:
- 7. It is only with <u>permission</u> that a person allowed to spend the night there.
 permission means:
- 8. Under the quarter-moon, which look pale becouse of the bright light of the city, he enjoyed watching and

listening to the birds.

pale means :

- 9. Many of the birds flew past the tower, while some slowed down and stop to rest for a while. slowed down means:
- 10. Teale was busy obserwing the birds when a <u>quard</u> on patrol found him.

quard means :

11. Looking down from the highest part of the building tourists can see cars running on the street <u>below</u> them.

below means :

- 12. Looking up they can see millions of stars above them above means:
- 13. And looking around they can see thousands of <u>lights</u>, many of them look brighter than the stars. <u>Lights</u> means:
- 14. And from there, people can <u>compare</u> the lights that are madexby God with that are made by man.
 <u>compare</u> means:
- 15. Often thousands of insects, and not <u>seldom</u> millions of them, cover the windows and walls of the tower.
 <u>seldom</u> means:

Name		*	•	•	•	•		•	•	•	•	•	
Reg.	Numb.												



Vocabulary Extending Test; the second reading

- 1. It was difficult for the new inhabitants of the New England to pass their first year <u>safely</u> and easily.
 <u>safely</u> means:
- 2. They were used to cold places, still the winter in the new country was too cold for many of them. winter means:
- Their life was full of danger and <u>hardships</u> but they continued to work hard.

hardships means :

4. In the homes or churches they said their <u>prayers</u> asking God to give protection.

prayers means :

5. When at last they have a very good harvest, their leader, the <u>governor</u> of New England, asked them to have a day of thanksgiving.

governor means :

6. They knew then that what they had prayed for was not in vain, not without result.

in vain means :

7. They believe more and more in praying, and in time of trouble they also had a day for fasting, during which they went without food.

fasting means :

8. And since that time American had always <u>celebrated</u> thanksgiving day, although not on the same day.

celebrate means :

- 9. Mrs. Hale sent many letters to governors and presidents, and her <u>articles</u> appeared in various magazines.
 <u>articles</u> means:
- 10. At last a <u>proclamation</u> was made by president Lincolp, deciding the fourth Thursday of November as the thanksgiving day.

proclamation means :

- 11. The president <u>proclaimed</u> the fourth Thursday of Nowember as the thanksgiving day in 1864.
 proclaimed means:
- 12. Actually, in 1799 President Washinton also <u>issued</u> a decicition, choosing November 26 as a day of general thanksgiving.

issued means :

13. Today Thanksgiving Day is really a <u>*pecial</u> day, during which stores, government offices, and schools are closed.

special means :

14. On that day American shave a <u>feast</u>, in which turkey is serve.

feast means :

15. In England, a thanksgiving ceremony took place and
was attended by the American Ambassador of the country.
ambassador means:

Name : Reg. Numb. :

APPENDIX D

Angket untuk Santri (siswa) kelas III Pesantren Modern IMMIM Tamalanrea Ujung Pandang

KETERANGAN

Daftar pertanyaan ini bertujuan untuk mengumpulkan data tentang hal-hal yang dianggap dapat mempengaruhi kemampuan santri dalam melakukan reading speed dan the extend of the vocabulary (membaca cepat dan perluasan kosa kata). Untuk itu peneliti menharapkan kepada anda agar memberikan jawaban yang sejujur-jujur nya sehingga hasil dari penelitian ini diperoleh de= ngan seobjektif mungkin. Kejujuran dan kesungguhan anda merupakan sumbangsih anda bagi penelitian ini, terlebih lagi bagi almamater anda.

PETUNJUK ANGKET :

- 1. Jawablah pertanyaan-pertanyaan berikut ini dengan memberikan tanda silang (X) pada hurup yang dianggap benar atau paling sesuai menurut anda.
- 2. Apabila terdapat pertanyaan yang dianggap kurang jelas atau belum dipahami supaya ditanyakan kepada peneliti.
- Tertarikkah anda mempelajari bahasa Inggris ?

 - a. sangat tertarik c. kurang tertarik
 - b. tertarik

- d. tidak tertarik
- 2. Pernahkah anda memperoleh informasi tentang speed

reading dan the extend of the vocabulary ?

a. pernah

b. tidak pernah

- 3. Bagaimana pendapat anda tentang reading speed dan the extend of the vocabulary ?
 - a. sangat penting untuk dipelajari
 - b. penting untuk dipelajari
 - c. agak penting untuk dipelajari
 - d. tidak penting untuk dipelajari
- 4. Tertarikkah anda untuk meningkatkan reading speed dan memperluas kosa kata yang anda miliki saat ini?

a. sangat tertarik c. agak tertarik

b. tertarik

- d. tidak tertarik
- 5. Apakah anda mempercayai bahwa reading speed dan vocabulary dapat ditingkatkan dan diperluas ?

a. Ya

b. tidak

6. Apakah anda sering membaca buku-buku, surat kabar, dan majalah-majalah yang berbahasa Inggris ?

a. sering

c. kadang-kadang

b. jarang

- d. tidak pernah
- 7. Berapa jamkah anda habiskan untuk membaca setiap hari ?
 - a. tiga jam atau lebih c. satu jam

b. dua jam

- d. setengah jam atau kurang
- 8. Berapa kalikah anda ke perpustakaan untuk membaca dalam seminggu ?
 - a. tiga kali atau lebih b. dua kali

d. tidak pernah
9. Sebelum anda membaca, apakah anda tahu atau menge
ti apa tujuan anda membaca ?
a. selalu c. kadang-kadang
b. jarang d. tidak pernah
10. Apakah anda sering mengalami kemunduran dalam mem
baca ?
a. sering c. kadang-kadang
b. jarang d. tidak pernah
11. Jika anda membaca, tahukah anda apa arti teks yang
yang sedang anda baca ?
a. Ya b. tidak .c. kadang-kadang ' Ya '
12. Apakah anda sering melatih kecepatan membaca dan
memperluas kosa kata yang anda miliki ?
a. sering c. kadang-kadang
b. jarang d. tidak pernah
13. Apakah anda mengerti pelajaran yang diterangkan o-
leh guru anda dalam bahasa Inggris ?
a. Ya b. Tidak c. kadang-kadang ' Ya '
14. Bagaimana usaha anda untuk meningkatkan reading
speed dan memperluas kosa kata yang anda miliki ?
15. Faktor-faktor apa sajakah yang menghambat anda un-
tuk meningkatkan reading speed dan memperluas kosa
kata yang anda miliki ?

c. satu kali